Author's response to reviews

Title: Proteome analysis of human gastric cardia adenocarcinoma by laser capture microdissection

Authors:

Yan Cheng (drchengyan@163.com)
Jun Zhang (jun3z@163.com)
Yang Li (iamdawenxi@163.com)
Yan Wang (wy@mti.xidian.edu.cn)
Jun Gong (jungong2003@163.com)

Version: 7 Date: 4 October 2007

Author's response to reviews: see over
Dear Dr. Scott Edmunds:

Enclosed herewith please find a manuscript entitled “Proteome analysis of human gastric cardia adenocarcinoma by laser capture microdissection” by Yan Cheng et al. We have revised the manuscript according to the comments of the editor. We have corrected the grammar mistakes of the manuscript, and have added all the authors to the reference. We marked the revised section by blue color. Please don’t hesitate to contact us if you have any problems or questions regarding the manuscript.

Sincerely yours,

Yan Cheng

Department of Gastroenterology
The Second affiliated Hospital
Xi’an Jiaotong University
North Road
Xi’an, Shaanxi
China
710004
Tel: (86)29-82325593
E-mail: drchengyan@163.com